\documentclass[a4paper,11pt]{amsart}

\newfont{\cyr}{wncyr10 scaled 1100}

\usepackage[left=2.7cm,right=2.7cm,top=3.5cm,bottom=3cm]{geometry}
\usepackage{amsthm,amssymb,amsmath,amsfonts,mathrsfs,amscd}
\usepackage[latin1]{inputenc}
\usepackage[all]{xy}
\usepackage{latexsym}
\usepackage{longtable}
\usepackage{color}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{question}[theorem]{Question}
\newtheorem{examplewr}[theorem]{Example}
\newtheorem{ass}[theorem]{Assumption}

\theoremstyle{remark}
\newtheorem{obswr}[theorem]{Observation}
\newtheorem{remarkwr}[theorem]{Remark}

\newenvironment{obs}{\begin{obswr}\begin{upshape}}{\end{upshape}\end{obswr}}
\newenvironment{remark}{\begin{remarkwr}\begin{upshape}}{\end{upshape}\end{remarkwr}}
\newenvironment{example}{\begin{examplewr}\begin{upshape}}{\end{upshape}\end{examplewr}}

\DeclareMathOperator{\shi}{ES}
\DeclareMathOperator{\dR}{\mathrm{dR}}
\DeclareMathOperator{\sym}{Sym}
\DeclareMathOperator{\Sym}{Sym}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\spec}{spec}
\DeclareMathOperator{\cris}{cris}
\DeclareMathOperator{\la}{la}
\DeclareMathOperator{\et}{et}
\DeclareMathOperator{\id}{Id}
\DeclareMathOperator{\reg}{reg}
\DeclareMathOperator{\pr}{pr}
\DeclareMathOperator{\cyc}{cyc}
\newcommand{\Lp}{{\mathscr{L}_p}}
\newcommand{\fc}{{\mathfrak c}}
\newcommand{\hV}{{\mathbb{V}}}
\newcommand{\Deltafp}{{^\flat\!\Delta'}}
\newcommand{\Deltaf}{{^\flat\!\Delta}}
\newcommand{\Deltap}{{\Delta'}}
\newcommand{\hphi}{\boldsymbol{\phi}}
\newcommand{\TT}{{\mathbb T}}
\newcommand{\uchi}{\mbox{\boldmath$\chi$}}
\newcommand{\testf}{\breve{f}}
\newcommand{\testg}{\breve{g}} 
\newcommand{\testh}{\breve{h}}
\newcommand{\testhf}{\breve{\hf}}
\newcommand{\testhg}{\breve{\hg}} 
\newcommand{\testhh}{\breve{\hh}}
\newcommand{\testG}{\breve{G}}
\newcommand{\testH}{\breve{H}}
\newcommand{\htf}{\breve{\mathbf{f}}}
\newcommand{\htg}{\breve{\mathbf{g}}}
\newcommand{\hth}{\breve{\mathbf {h}}}
\newcommand{\LLambda}{\Lambda^{\!_{(N)}}}
\newcommand{\cF}{\mathcal F}
\newcommand{\cH}{\mathcal H}
\newcommand{\cC}{\mathcal C}
\newcommand{\V}{\mathcal V}
\newcommand{\cE}{\mathcal E}
\newcommand{\cU}{\mathcal U}
\newcommand{\cW}{\mathcal W}
\newcommand{\cA}{\mathcal A}
\newcommand{\g}{\gamma}
\newcommand{\G}{\Gamma}
\newcommand{\Gap}{\Gamma_0^D(p M)}
\newcommand{\Ga}{\Gamma_0^D(M)}
\newcommand{\hGa}{\hat{\Gamma }_0(M)}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\F}{\mathbb{F}}
\newcommand{\K}{\mathbb{K}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\I}{\mathbb{I}}
\newcommand{\Ql}{\Q_{\ell}}
\newcommand{\GQ}{G_{\Q}}
\newcommand{\PP}{\mathbb{P}}
\newcommand{\Sel}{\mathrm{Sel}}
\newcommand{\Sha}{\mathrm{Sha}}
\newcommand{\Gal}{\mathrm{Gal\,}}
\newcommand{\GL}{\mathrm{GL}}
\newcommand{\Cl}{\mathrm{Cl}}
\newcommand{\Tr}{\mathrm{Tr}}
\newcommand{\Div}{\mathrm{Div}}
\newcommand{\Fil}{\mathrm{Fil}}
\newcommand{\rank}{\mathrm{rank}}
\newcommand{\cond}{\mathrm{cond}}
\newcommand{\Frob}{\mathrm{Frob}}
\newcommand{\End}{\mathrm{End}}
\newcommand{\Jac}{\mathrm{Jac}}
\newcommand{\disc}{\mathrm{disc}}
\newcommand{\Id}{\mathrm{Id}}
\newcommand{\MS}{\mathrm{MS}}
\newcommand{\Aut}{\mathrm{Aut}}
\newcommand{\Fr}{\mathrm{Fr}}
\newcommand{\diag}{\mathrm{diag}}
\newcommand{\Up}{\mathrm{U}_p}
\newcommand{\ord}{{\mathrm{ord}}}
\newfont{\gotip}{eufb10 at 12pt}
\newcommand{\gn}{\mbox{\gotip n}}
\newcommand{\gm}{\mbox{\gotip m}}
\newcommand{\ga}{\mbox{\gotip a}}
\newcommand{\gb}{\mbox{\gotip b}}
\newcommand{\gf}{\mbox{\gotip f}}
\newcommand{\gee}{\mbox{\gotip e}}
\newcommand{\go}{{\mathcal O}_{A_f}}
\newcommand{\OAf}{{\mathcal O}_{A_f}}
\newcommand{\gp}{\mbox{\gotip p}}
\newcommand{\gL}{\mbox{\gotip L}}
\newcommand{\gd}{\mbox{\gotip d}}
\newcommand{\cO}{{\mathcal O}}
\newcommand{\cM}{{\mathcal M}}
\newcommand{\cP}{{\mathcal P}}
\newcommand{\cL}{{\mathcal L}}
\newcommand{\cG}{{\mathcal G}}
\newcommand{\GG}{{\mathcal G}}
\newcommand{\cD}{{\mathcal D}}
\newcommand{\om}{{\omega}}
\newcommand{\ra}{\rightarrow}
\newcommand{\lra}{\longrightarrow}
\newcommand{\hra}{\hookrightarrow}
\newcommand{\Norm}{\mathrm{N}}
\newcommand{\NS}{\mathrm{NS}}
\newcommand{\SL}{{\mathrm {SL}}}
\newcommand{\SO}{{\mathrm {SO}}}
\newcommand{\n}{{\mathrm{n}}}
\newcommand{\Pic}{{\mathrm{Pic}}}
\newcommand{\CH}{{\mathrm{CH}}}
\newcommand{\AJ}{{\mathrm{AJ}}}
\newcommand{\tr}{{\mathrm{tr}}}
\newcommand{\si}{\sigma}
\newcommand{\eps}{\epsilon}
\newcommand{\ta}{{\tau}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\M}{{\mathrm{M}}}
\newcommand{\HC}{\mathcal F_{\rm har}}
\newcommand{\PGL}{{\mathrm{PGL}}}
\newcommand{\Supp}{\mathrm{Supp}}
\newcommand{\ep}{\varepsilon}
\newcommand{\cX}{\mathcal X}
\newcommand{\hf}{{\mathbf{f}}}
\newcommand{\hg}{{\mathbf{g}}}
\newcommand{\hh}{{\mathbf h}}
\newcommand{\ha}{{\mathbf a}}
\newcommand{\hB}{{\mathbf B}}
\newcommand{\VV}{{\mathbf V}}
\newcommand{\DD}{{\mathbf D}}
\newcommand{\homega}{{\mathbf \omega}}
\newcommand{\heta}{{\mathbf \eta}}
\newcommand{\hkappa}{{\mathbf \kappa}}
\DeclareMathOperator{\cts}{cts}
\newcommand{\cl}{{\mathrm{cl}}}
\def \mint {\times \hskip -1.1em \int}
%\newcommand{\mint}{\times\!\!\!\!\!\hskip-0.3em\int}
\newcommand{\SP}{\vspace{0.5cm}}
\newcommand{\T}{\mathbb T}
\newcommand{\new}{\mathrm{new}}
\DeclareMathOperator{\Hom}{Hom} \DeclareMathOperator{\Meas}{Meas}
\newcommand{\X}{\mathbb X}
\newcommand{\Y}{\mathbb Y}
\newcommand{\U}{\mathbb U}
\newcommand{\Rat}{{\rm Rat}\bigl(\PP^1(\Q_p),\C_p\bigr)}
\newcommand{\Ker}{{\mathrm Ker}}
\newcommand{\res}{\mathrm{res}}
\DeclareMathOperator{\Ind}{Ind}
\newcommand{\Hida}{{\mathcal H}}
\newcommand{\rom}{\mathrm}
\newcommand{\fr}{\mathfrak}
%\newcommand{\cl }{\mathcal}
\DeclareMathOperator{\an}{an}
\DeclareMathOperator{\lcm}{lcm}
\newcommand{\testgamma}{\breve{\gamma}}
\newcommand{\longmono}{\mbox{$\lhook\joinrel\longrightarrow$}}
\newcommand{\longleftmono}{\mbox{$\longleftarrow\joinrel\rhook$}}
\newcommand{\longepi}{\mbox{$\relbar\joinrel\twoheadrightarrow$}}
\newcommand{\mat}[4]{\left(\begin{array}{cc}#1&#2\\#3&#4\end{array}\right)}
\newcommand{\smallmat}[4]{\bigl(\begin{smallmatrix}#1&#2\\#3&#4\end{smallmatrix}\bigr)}
\newcommand{\dirlim}{\mathop{\varinjlim}\limits}
\newcommand{\invlim}{\mathop{\varprojlim}\limits}
\newcommand{\D}{\mathbb D}
\newcommand{\ka}{{\kappa}}
\newcommand{\lam}{{\lambda}}
\newcommand{\cEul}{\mathcal Eul}
\newcommand{\barchi}{{\overline{\chi}}}
\newcommand{\gak}{{\ga_{\mathrm K}}}
\newcommand{\gek}{{\gee_{\mathrm K}}}
\newcommand{\gfk}{{\gf_{\mathrm K}}}
\newcommand{\gah}{{\ga_{\mathrm GH}}}
\newcommand{\geh}{{\gee_{\mathrm GH}}}
\newcommand{\gfh}{{\gf_{\mathrm GH}}}
\newcommand{\gab}{{\ga_{\mathrm BDP}}}
\newcommand{\geb}{{\gee_{\mathrm BDP}}}
\newcommand{\gfb}{{\gf_{\mathrm BDP}}}
\newcommand{\gfp}{{\gf_{\mathrm{Pet}}}}
\newcommand{\gap}{{\ga_{\mathrm{Pet}}}}
\newcommand{\gar}{{\ga_{\mathrm HR}}}
\newcommand{\ger}{{\gee_{\mathrm HR}}}
\newcommand{\gfr}{{\gf_{\mathrm HR}}}
\newcommand{\A}{{\mathbb A}}
\newcommand{\Zp}{{\Z_p}}
\newcommand{\Qp}{{\Q_p}}
\newcommand{\Qpbar}{{\overline{\Q}_p}}
\newcommand{\Cp}{{\C_p}}
\newcommand{\barQ}{\bar\Q}
\newcommand{\bp}{{\bold p}}
\newcommand{\cont}{{\mathcal C}}
\newcommand{\BM}{{\mathcal{BM}}}
\newcommand{\N}{{\mathbb N}}
\newcommand{\Spf}{{\mathrm{Spf}}}




\include{thebibliography}






%COMMENTS ON THE CORRECTIONS POINT BY POINT:

%(1)			Still to do, just want a clue on how it is better to do in order to solve it: Eichler-Shimura provide E, not just an isogeny class, but an actual elliptic curve, that I know, the "Strong Weil Parametrization", no? Then it defines of course an isogeny class, but our formula just works for the optimal quotient (we should add somewhere the degree of the isogeny, probably squared, to get the result for the other class). In case everything I say is fine, I add it to the comments for the referee.
%(2)		DONE (after the displayed equation)
%(3)		DONE.
%(4)		DONE (kept the notation of page 7 because it is the norm, not the norm Hecke character)
%(5)	(a)	DONE.
%	(b)		I think that this is a good point. I asked you a lot lot lot of time ago something related to that and you told me that the Heegner hypothesis took care of everything here, but I am not sure 100% (or I do not see why). In the paper of Gross, Heegner points on X0(N), at the section 11 (page 92) he explains how one should assume that:
%					a_q(f) = -a_q(\psi),   for all q | (D_K, N_f)
% otherwise the Heegner point is trivial. So our hypotheses work fine, but only after adding this detail. This is equivalent to impose a condition on the local signs at these primes being positive (cf. always Gross, page 101 (section 22)
%	(c)	DONE (directly in equation 1)
%(6)		DONE (its mathfrak N, I corrected the other)
%(7)		DONE (H was already defined 4-5 lines before equation (1). I added the explanation of G just after equation (2))
%(8)		DONE (I put the definition in page 2, just before the first occurrence of \psi')
%(9)		DONE.
%(10)	DONE.
%(11)		To explain to the referee: it is coherent because we took the conductors of the Hecke characters equals to 1 and we were left with N_f=D_K
%(12)	DONE (I put everywhere N_f)
%(13)	DONE (I deleted that, the sentence is even better without, I think, although I think that throughout was fine)
%(14)	DONE (was just a mistake)
%(15)		To explain to the referee: well, yes! Or am I mistaken?
%(16)	From what I have seen in proposition 2.6 of DLR, I would say yes. we take the p-depletion of f, so the fact that is new or not, should not count. Anyway it seems to me that the argument made in loc. cit. perfectly works when we substitute everything with the respective p-stabilisations, am I right?
%(17)	DONE (I put N_f)
%(18)	DONE (I modified the section and recalled everywhere possible the weight of f)
%(19)	DONE.
%(20)	DONE (I added the classicality hypotheses, in red in the introduction, to check)
%(21)		To explain to the referee: it comes from equations (6), from f_{BDP}, which is defined in BDP13, theorem 4.6. There it is obtained as a result of computations of local integrals at primes p | (D_K,N_f)











\begin{document}
\begin{thebibliography}{99}


\bibitem[ARS]{ARS}
A. Agashe, K. Ribet, W. Stein, {\em The Manin constant}, Pure Appl. Math. Q., Issue 2, (2006), 617--636

\bibitem[BeDi]{BeDi} J. Bella\"iche, M. Dimitrov, {\em On the eigencurve at classical weight one points}, Duke Math. J. 165, Issue 2 (2016), 245--266, available at {\tt http://math.univ-lille1.fr/$\sim$mladen/}

\bibitem[BD1]{BD1}
M. Bertolini, H. Darmon, {\em Kato's Euler system and rational points on elliptic curves I: a $p$-adic Beilinson formula }, Israel Journal of Math., {\bf 199}, Issue 1, January 2014, 163--188

%\bibitem[BD2]{BD2}
%M. Bertolini, H. Darmon, {\em Perrin-Riou's conjecture for elliptic curves}, in progress.


%\bibitem[BD2]{BD5}
%M. Bertolini, H.~Darmon, 

\bibitem[BD3]{BD}
M. Bertolini, H.~Darmon, {\em The rationality of Stark-Heegner points over genus fields of real quadratic fields}, Annals of math., {\bf 170} (2009), 343--369

\bibitem[BDR1]{BDR1}
M. Bertolini, H. Darmon, V. Rotger, {\em Beilinson-Flach elements and Euler systems I: syntomic regulators and $p$-adic Rankin L-series}, Journal of Algebraic Geometry {\bf 24}, (2015), 355--378

\bibitem[BDP13]{BDP1}
 M. Bertolini, H. Darmon, K. Prasanna, {\em Generalised Heegner cycles and p-adic
Rankin L-series} (with an appendix by Brian Conrad), Duke Math. J. {\bf 162} (2013), no. 6, 1033--1148. 

\bibitem[Ca1]{Cas1}
F. Castell\`{a}, {\em On the $p$-adic variation of Heegner points}, Journal de l'Inst. Math. de Jussieu, to appear

\bibitem[Ca2]{Cas2}
F. Castell\`{a}, {\em On the exceptional specializations of big Heegner points}, Journal de l'Inst. Math. de Jussieu, to appear

\bibitem[CR]{CR}
D. Casazza, V. Rotger, {\em Stark points and p-adic iterated integrals}, Ramanujan Journal of Math., to appear 

\bibitem[Dar]{darmon-hpxh}
H.~Darmon, {\em  Integration on $\cH_p\times \cH$ and arithmetic applications}. 
Ann. of Math. (2) 154 (2001), no. 3, 589--639.

\bibitem[DLR]{DLR}
H.~Darmon, A. Lauder, V. Rotger, {\em Stark points and p-adic iterated integrals attached to modular forms of weight one}, Forum of Math. Pi, {\bf 3}, e8, 95 pages


\bibitem[DR1]{DR1}
H. Darmon, V. Rotger, 
{\em Diagonal cycles and Euler 
systems I: a $p$-adic Gross-Zagier formula},  Annales Scientifiques de l'Ecole Normale Sup\'erieure {\bf 47}, n. 4 (2014), 779--832.

\bibitem[DR2]{DR2}
H.~Darmon and V.~Rotger, {\em 
Diagonal cycles and Euler systems II:   the Birch and Swinnerton-Dyer conjecture for Hasse-Weil-Artin $L$-series}, Journal of the Amer. Math. Soc., to appear 

\bibitem[DR4]{DR4}
H.~Darmon and V.~Rotger, {\em A Kolyvagin theorem in rank $2$}, in progress. 

\bibitem[DG]{DG}
M. Dimitrov, E. Ghate, {\em On classical weight one forms in Hida families}, J. Th\'eorie des Nombres de Bordeaux, 24, Issue 3 (2012), 669-690.

 \bibitem[Gar]{Gar}
P. Garrett, {\em Decomposition of Eisenstein series:\,Rankin triple products}, Annals Math. {\bf 125} (1987), 209-235.

%\bibitem[GS]{GS}
%R. Greenberg, G. Stevens, ...

\bibitem[Gro]{Gro}
B. Gross, {\em Heegner points on $X_0(N)$}, chapter 5 of {\em Heegner points and Rankin L-series}, Cambridge University Press (Mathematical Sciences Research Institute Publications), (2004)  doi:10.1017/CBO9780511756375

\bibitem[Hi86]{Hi86}
H. Hida, {\em Galois representations into $\GL_2(\Z_p[[X]])$ attached to ordinary cusp forms}, Invent. Math., {\bf 85}, no. 3, (1986), 545-613

\bibitem[Hi93]{Hi93}
H. Hida, Elementary theory of $L$-functions and Eisenstein series, London Mathematical Society Student Texts {\bf 26}, 1993.


\bibitem[Ho07]{How}
B. Howard, {\em Variation of Heegner points in Hida families}, Invent. Math. {\bf 167}, No.1 (2007), 91-128

\bibitem[Ja88]{Ja} U. Jannsen, {\em Continuous \'etale cohomology}, Math. Annalen {\bf 280} (1988), 207--245.

\bibitem[Ka10]{Kani2}
E. Kani, {\em Binary theta series and modular forms with complex multiplication}, Int. J. Number Theory, {\bf 10} n.4 (2010), 1025--1042

\bibitem[Ka76]{Katz1}
N. M. Katz, {\em $p$-adic interpolation of real analytic Eisenstein series}, Annals Math. {\bf 104}, no.\,3 (1976), 459-571.

\bibitem[Ka78]{Katz2}
N. M. Katz, {\em $p$-adic $L$-functions for CM fields}, Invent. Math. {\bf 49} (1978), 199--297.

%\bibitem[Ki]{Ki}
%G. Kings,

%\bibitem[KLZ]{KLZ}
%G. Kings, D. Loeffler, S. Zerbes, 

\bibitem[KZ84]{KZ} W. Kohnen, D. Zagier, {\em Modular forms with rational periods}, in
"Modular forms" (Durham, 1983), 197--249, Ellis Horwood Ser. Math.
Appl.: Statist. Oper. Res., Horwood, Chichester, 1984.

\bibitem[LRdV]{LRdV} M. Longo, V. Rotger, C. de Vera-Piquero, {\em Heegner points on Hijikata-Pizer-Shemanske curves},  Publ. Matem\`{a}tiques, to appear.

\bibitem[PR93]{PR} B. Perrin-Riou, {\em Fonctions $L$ p-adiques d'une courbe elliptique et points rationelles}. Ann. Inst. Fourier, Grenoble {\bf 43}, n.4 (1993), 945--995.

\bibitem[Pr90]{Pr} D. Prasad, {\em Trilinear forms for representations of $\GL(2)$ and local epsilon factors}, Compositio Math. {\bf 75} (1990), 1--46.

\bibitem[Wi88]{W} A. Wiles, {\em On ordinary $\Lambda$-adic representations associated to modular forms},
 Inv.\,Math. {\bf 94} (1988), 529-573.

\bibitem[Wo1]{Wood}
 M. Woodbury. {\em Trilinear forms and subconvexity of the triple product $L$-function}, submitted.

\bibitem[Wo2]{Wood2} 
M. Woodbury. {\em On the triple product formula: real local calculations}, submitted.


\end{thebibliography}


\title[On the elliptic Stark conjecture]{On the elliptic Stark  conjecture at primes of multiplicative reduction}

\author{Daniele Casazza and Victor Rotger}


\begin{abstract}
 In \cite{DLR}, Darmon, Lauder and Rotger formulated a $p$-adic elliptic Stark conjecture for the twist of an elliptic curve $E/\Q$ by the self-dual tensor product $\rho_1\otimes \rho_2$ of two odd and two-dimensional Artin representations. The authors of loc.\,cit.\,provided abundant numerical evidence and proved the conjecture in the special setting where $p$ is a prime of good reduction for $E$ and $\rho_1$ and $\rho_2$ are induced from finite order characters $\psi_g$, $\psi_h$ of the same imaginary quadratic field. The key step in their proof is a factorization of one-variable $p$-adic $L$-functions, where $\psi_g$ varies in a $p$-adic family of Hecke characters.
 
The main goal of this article is to prove a new case of the conjecture, placing ourselves in the setting where $p$ is a prime of multiplicative reduction for $E$. In order to achieve our theorem, we need to work with {\em two-variable} $p$-adic $L$-functions, where the weight $2$ cusp form associated with $E$ also moves independently along a Hida family. Our main result then follows from a factorization of $p$-adic $L$-series extending to two-variables the one obtained in \cite{DLR}. On the way we also generalize to our setting the results obtained in \cite{CR}.
\end{abstract}


%========== Informations about the AUTHORS ==========
\address{D. C.: ICMAT - CSIC, C. de Nicol\'as Cabrera 13-15, 28049 Madrid, Spain}
\email{daniele.casazza@icmat.es}
\address{V. R.: Departament de Matem\`{a}tica Aplicada II, Universitat Polit\`{e}cnica de Catalunya, C. Jordi Girona 1-3, 08034 Barcelona, Spain}
\email{victor.rotger@upc.edu}


%\subjclass{11G18, 14G35}


\maketitle

\tableofcontents

\section{Introduction}\label{sec1}

Fix a prime number $p\geq 3$ and three positive integers $N_f, N_g, N_h$ such that  $p\nmid N_f N_g N_h$. Set $N=\mathrm{lcm}(N_f,N_g,N_h)$ and let $\chi: (\Z/N\Z)^\times \ra \C^\times$ be a Dirichlet character. Let
$$f\in S_2(p N_f), \quad g\in M_1(N_g,\bar\chi), \quad h\in M_1(N_h,\chi)$$ be a triple of newforms of weights $(2,1,1)$, levels $(p N_f,N_g,N_h)$ and nebentype  characters $(1,\bar\chi,\chi)$. 

Assume for simplicity that the Fourier coefficients of $f$ are rational and let $E/\Q$ be an elliptic curve in the isogeny class associated with $f$ by Eichler-Shimura. Let also $\varrho_g$, $\varrho_h$ denote the odd, two-dimensional Artin representations associated by Deligne-Serre to $g$ and $h$. This paper focuses on the arithmetic of the elliptic curve $E$ twisted by the four-dimensional Artin representation $$\varrho_{gh}:=\varrho_g \otimes\varrho_h,$$ which is self-dual because the nebentype characters of $g$ and $h$ are inverse to each other. 

Let $L(E,\varrho_{gh},s)$ denote the Hasse-Weil-Artin $L$-series associated with the compatible system of Galois representations afforded by the twist of $E$ by $\varrho_{gh}$. A direct consequence of the Eichler-Shimura and Deligne-Serre constructions is the identity
$$
L(E,\varrho_{gh},s) = L(f,g,h,s),
$$
which recasts the $L$-function associated with the pair $(E,\varrho_{gh})$ as Garrett's \cite{Gar} triple-product $L$-function attached to the triple of eigenforms $(f,g,h)$.


Assuming that $(p N_f,N_g N_h)=1$, the global sign of the functional equation of  $L(E,\varrho_{gh},s)$ is $+1$. We refer e.g.\,to \cite{Pr} for a proof of this simple fact and for similar assertions under more relaxed hypotheses. 

Assume further that $L(E,\varrho_{gh},s)$ vanishes at the central critical point $s=1$. It follows that the order of vanishing of $L(E,\varrho_{gh},s)$ at $s=1$ is even and hence at least $2$.

In \cite{DLR}, Darmon, Lauder and one of the authors of this note introduced certain {\em $p$-adic iterated integrals} attached to the triple $(f,g,h)$, which they view as $p$-adic avatars of the leading term of $L(E,\varrho_{gh},s)$ at $s=1$. These iterated integrals are defined as linear forms
$$
\begin{matrix}
S_2(Np)_L[f] \times  M_1(Np,\chi)_L^\vee[g_\alpha]\times M_1(Np,\chi)_L[h] & \lra \qquad \C_p \\ & \\
(\testf,  \testgamma, \testh)  & \mapsto  \qquad  \int_{\testgamma} \! \testf \cdot \testh
\end{matrix}
$$
on the space of triples of test vectors of level $Np$ on which the good Hecke operators act with the same system of eigenvalues as $f$, $g_\alpha$ and $h$ respectively. Here $g_\alpha$ is a choice of (necessarily ordinary) $p$-stabilization of $g$, on which $U_p$ acts with eigenvalue $\alpha$.

 Under our hypotheses and an additional assumption referred to as Hypothesis C in \cite{DLR} on the local geometry of the points corresponding to the ordinary stabilisations of $g$ in the eigencurve,  the authors of loc.\,cit.\,propose a conjectural formula for these iterated integrals which may be regarded as a $p$-adic avatar of a putative Gross-Zagier formula in rank $2$. 

More precisely, these iterated integrals are expected to vanish systematically when the second derivative $L''(E,\varrho_{gh},1)$ vanishes; if instead  $L''(E,\varrho_{gh},1)\ne 0$, the main conjecture of \cite{DLR} predicts that the values of the iterated integrals attached to $(f,g,h)$ encode the logarithm of a suitable Gross-Stark unit and a $2\times 2$ $p$-adic regulator  involving  the formal group logarithms of certain {\em Stark points} in the Mordell-Weil group of $E$ over the number field cut out by $\varrho_{gh}$. This conjecture was christened in \cite{DLR} as the {\em elliptic Stark conjecture}.

The main theorem of \cite[Ch.\,2 \& 3]{DLR} shows that the elliptic Stark conjecture holds when $p$ is a prime of good reduction for $E$, while $g$ and $h$ are binary theta series attached to a common imaginary quadratic field $K$ in which $p$ splits. This is done by recasting both sides of the conjectural formula, that is to say $p$-adic iterated integrals on one side and elliptic units and Heegner points on the other, as values of various $p$-adic $L$-functions at points lying outside their region of interpolation. 

The first main theorem of this article extends the above result to the case where $p$ is a prime of multiplicative reduction of $E$. 
In order to state the result precisely, let us briefly review the ingredients appearing in its formulation.

Fix throughout embeddings $\barQ \subset \C$ and $\barQ\subset \Cp$ so that every number field shall be regarded without further mention as a subfield of both $\C$ and $\C_p$. 

Let $K$ be an imaginary quadratic field of discriminant $-D_K$ and let us denote by $\chi_K$ the quadratic Dirichlet character associated with the extension $K/\Q$. Assume that $p$ splits in $K$, i.e.\,$\chi_K(p)=1$. 

Let $g$ and $h$ be theta series associated with two finite order characters $\psi_g, \psi_h: G_K \lra L^\times$ of $K$ of conductors $\mathfrak c_g$ and $\mathfrak c_h  \subseteq \cO_K$ respectively, with values in a finite extension $L$ of $\Q$. They are weight one forms of levels $D_K \Norm_{K/\Q}(\mathfrak c_g)$ and $D_K \Norm_{K/\Q}(\mathfrak c_h)$, having nebentype characters $\chi_g$ and $\chi_h$ respectively. 

Define $\psi_g': G_K \lra L^\times$ to be the $\Gal(K/\Q)$-conjugate of the character $\psi_g$, namely the character given by the rule $\psi_g'(\sigma) = \psi_g(\tau \sigma \tau^{-1})$ for any $\sigma \in G_K$ and any $\tau \in G_{\Q} \setminus G_K$. 

We assume that $g$ satisfies the classicality hypotheses of \cite{DLR} at the prime $p$. As explained in loc. cit, this is automatically satisfied
\begin{itemize}
\item when $\psi_g/\psi_g' = 1$ (i.e. $g$ is Eisenstein);
\item when $\psi_g/\psi_g' \ne 1$ (i.e. $g$ is cuspidal), if we impose $\psi(\wp) \ne \psi'(\wp)$ and that $\rho_g$ is not induced from a character of a real quadratic field in which $p$ splits.
\end{itemize}


As above, we assume that $\chi_h=\chi_g^{-1} (= \chi)$, which implies that 
the characters $$\psi_1= \psi_g \psi_h \quad \mbox{ and } \quad \psi_2= \psi_g \psi'_h$$  are ring class characters of $K$, associated with  orders $\cO_{c_1}$  and $\cO_{c_2}$ in $\cO_K$ 
 of conductors $c_1$ and $c_2$ respectively. Let $H$ denote the ring class field associated with the order $\cO_c$ in $K$ of conductor $c:=\mathrm{lcm}(c_1,c_2)$ of $K$,  which contains the fields cut out by $\psi_1$ and $\psi_2$. 
 
A simple exercise shows that the Artin representation $\varrho_{gh} = \varrho_g\otimes \varrho_h$ decomposes as
$$
\varrho_{gh} = \rho_1 \oplus \rho_2 =  V_{\psi_1} \oplus V_{\psi_2},
$$
and the Artin formalism in turn implies that there is a factorisation of $L$-series
\begin{equation}
\label{eqn:dihedral-decomp}
 L(E,\varrho_{gh},s) = L(E,\rho_1,s)\cdot L(E, \rho_2,s) = L(E/K,\psi_1,s)\cdot L(E/K,\psi_2,s).
 \end{equation}
 
Recall that  $N=\mathrm{lcm}(N_f,N_g,N_h) = \mathrm{lcm}(N_f,D_K \Norm_{K/\Q}(\mathfrak c_g),D_K \Norm_{K/\Q}(\mathfrak c_h))$. Assume in this introduction that $N$ is square-free, $(N_f D_K,c)=1$ and the following assumption holds (which is exactly the one imposed in \cite[\S 4.1]{BDP1}):


%If we assume that $\gcd(pN_f, \mathfrak c_g \mathfrak c_h) =1$, then  the local signs of $L(E/K,\psi_1,s)$ and of $L(E/K,\psi_2,s)$ are all equal for all places of $K$ above any place $v\mid N \infty$ of %$\Q$, and hence $\varepsilon_v(E,\varrho_{gh}) = (\pm 1)^2=1.$ This automatically implies that Hypothesis A in \cite{DLR} holds.




\medskip\noindent
{\bf Heegner hypothesis:} All prime divisors of $N_f$ split or ramify in $K$, and the local signs at primes dividing $\mathrm{gcd}(N_f,D_K)$ of  $L(E,\rho_1,s)$ and $L(E,\rho_2,s)$ are $+1$.

\medskip\noindent

We refer to \cite[\S 11 and 22]{Gro} for more details about this hypothesis. As shown in loc.\,cit., this assumption automatically implies that the local signs of $L(E,\rho_1,s)$ and $L(E,\rho_2,s)$  are $+1$ at {\em all} finite primes. Since the local sign at $\infty$ of  $L(E,\rho_i,s)$ is $-1$, this assumption implies that the global signs of the functional equations satisfied by $L(E/K,\psi_1,s)$ and $L(E/K,\psi_2,s)$ are $-1$,  hence the analytic ranks  $r_{\an}(E,\varrho_{\psi_1})$ and $r_{\an}(E,\varrho_{\psi_2})$ are both odd. 

It also follows from \eqref{eqn:dihedral-decomp} that $L(E,\varrho_{gh},1)=0$ and thus Hypothesis $B$ of \cite{DLR} is satisfied. Moreover, since the local sign at any place of $L(E,\varrho_{gh},s)$ is the product of the local signs of  $L(E/K,\psi_1,s)$ and of $L(E/K,\psi_2,s)$, it follows that Hypothesis A in \cite{DLR} holds.

Finally, the Heegner hypothesis also implies that there exists an integral ideal $\mathfrak N$ in $\cO_K$ such that $\Norm_{K/\Q}(\mathfrak N)=N_f$. We fix such an ideal for the remainder of the article.

 Let $X$ denote the modular curve associated in \cite[Theorem A]{LRdV} to the triple $(E,K,c)$ and let
\begin{equation*}
\pi_E: X\rightarrow E
\end{equation*}
be a modular parametrization of $E$. 


Under our assumptions we let $X(H)_{c} \subset X(H)$ denote the set of Heegner points attached to the order $\cO_c$. After fixing a choice of $P\in X(H)_{c}$, any ring class character $\psi$ of conductor $c$ gives rise to a degree zero $H$-rational divisor on $X$ by setting
\begin{equation}\label{eqn:choice-of-P}
	 D_\psi := \left\{ \begin{array}{ll}
		 \sum_{\sigma\in G} \psi^{-1}(\sigma) P^\sigma, & \mbox{ if  } \psi \ne 1, \\
		  \sum_{\sigma\in G}  P^\sigma -  \#G\cdot \infty & \mbox{ if } \psi=1,
	  \end{array} \right.
\end{equation}
where $\infty$ is the cusp at infinity and $G := \Gal(H/K)$. The {\em Heegner point attached to $\psi$} is then defined to be
 $$ P_\psi := \pi_E(D_\psi) \in  E(H)\otimes L.$$
 
 

Set $\psi_0 := \psi_g/\psi_g'$ and note that this is again a ring class character of $K$. Fix an elliptic unit $u$ in the field $H_g$ cut out by $\psi_0$ as in \cite[\S 3.1]{DLR} and define
   \begin{equation} \nonumber
%% \label{ug-general}
u_{\psi_0} = 
\sum_{\sigma \in G}  \psi_0^{-1}(\sigma) u^{\sigma} \in (\cO_{H_g}^\times)\otimes L.
\end{equation}

Let $\omega_E$ denote the N\'eron differential $1$-form of $E$. The pull-back of this form to $X$ is a non-zero rational multiple of $\omega_f$, the canonical differential $1$-form on $X$ associated with $f$. Hence $\pi_E^*(\omega_E) = c_E\cdot \omega_f$ for some $c_E\in \Q^\times$, usually referred to as the Manin constant of the modular parametrization. We refer to \cite{ARS} for a survey of known results about the Manin constant.


Fix  embeddings of $H_g \subseteq H$ into $\bar\Q_p \subset \C_p$ and let $H_{p}$ denote the closure of $H$ in $\C_p$. Let
$$
\log_{E,p}: E(H_p) \, \lra \, H_p
$$
denote the formal group logarithm associated with $\omega_E$. Let also
$$
\log_p: \cO_{H_{p}}^\times \,\lra \,\cO_{H_{p}}
$$
denote the usual $p$-adic logarithm. The main theorem of this note is the following.

\begin{theorem}
\label{main-theta} 
If either $L'(E/K,\psi_1,1)$ or $L'(E/K,\psi_2,1)$ vanishes,  the iterated integrals attached
to the triple $(f,g,h)$ vanish identically. 

Otherwise,
there exists a finite extension $L$ of $\Q(\psi_g, \psi_h)$, a scalar $\lambda \in L^\times$  and test vectors 
$$(\testf,  \testgamma_{g_\alpha}, \testh)  \in S_2(Np)_L[f] \times  M_1(Np,\chi)_L^\vee[g_\alpha]\times M_1(Np,\chi)_L[h] $$  
for which
\begin{equation*}
 \int_{\testgamma_{g_\alpha}} \! \testf \cdot \testh  =  \lambda \cdot \frac{\log_{E,p}(P_{\psi_1}) \log_{E,p}(P_{\psi_2})}{\log_p(u_{\psi_0})}.
\end{equation*}
\end{theorem}

%The proof of this theorem is given in \S \ref{proof1}. 
The second main result of this note is a precise formula for the value of $\lambda$ appearing in the above statement, in a particular setting.
\begin{theorem}\label{thm-lambda} 
Assume  $N_f = D_K$ and $g=h=\mathrm{Eis}(1,\chi_K)$ are the Eisenstein series associated with the pair of Dirichlet characters $(1,\chi_K)$. Let $h_K=|\Pic(\cO_K)|$ and $g_K=|\Pic(\cO_K)/\Pic^2(\cO_K)|$ denote the class number and genus number of $K$. Then
	\[
		\lambda = \frac{(p-a_p(f) \psi(\wp))^2}{p \cdot c_E^2}\cdot \frac{\lambda_0 }{h_K g_K}
	\]
	where:
	\[
		\lambda_0 = \begin{cases} \frac{1}{p-1}   & \quad \mbox{if } \psi^2=1, \mbox{ that is to say, if } g \mbox{ is Eisenstein} \\
\frac{12}{p-(p+1)\psi^2(\bar \wp)+\psi^4(\bar \wp) } & \quad \mbox{if } \psi^2\ne 1, \mbox{ that is to say, if } g \mbox{ is cuspidal.}
\end{cases}	
	\]
\end{theorem}

%{\color{blue} \begin{remark}
%	Notice that because we are assuming conditions on the sign of the functional equations, if $q \mid %D_K$ and $\mathfrak q$ is the prime above $q$ in $K$, then $\psi(\mathfrak q) = -a_q(f)$ (cf. %\cite[Lemma 11.1]{Gro}). Hence $\psi((\sqrt{-D}))= (-1)^{\#q\mid D} a_D(f)$.
%\end{remark}}

The proof of these results is not a straight-forward generalization of \cite{DLR} and \cite{CR} respectively, mainly due to the following reasons:

\begin{enumerate}
\item[I.] One of the main ingredients in the proof of Theorems \ref{main-theta} and \ref{thm-lambda} is the main result of Bertolini, Darmon and Prasanna in \cite{BDP1}, which is proved in loc.\,cit.\,under the assumption that $p$ is a prime of good (although not necessarily ordinary) reduction.

\item[II.] Another crucial ingredient in our proofs is the use of various $p$-adic $L$-functions, like Garrett-Hida's $p$-adic $L$-function constructed in \cite{DR1}, \cite[Ch.\,2]{DLR} and Bertolini-Darmon-Prasanna's $p$-adic $L$-function described in \cite[\S 5]{BDP1}. In the interpolation formulae relating these functions to critical values of classical $L$-series there appear certain auxiliary constants as for instance the ones denoted $C_v$ in \cite[Proposition 2.1]{DLR} and also the one denoted $\omega(f, \psi)$ in \cite[(5.1.11)]{BDP1}.  When $p$ is a prime of bad reduction, it is a rather subtle problem to analyze the $p$-adic valuation of these constants and its variation along Hida families of modular forms.

\item[III.] In order to find an explicit formula for the scalar $\lambda$ appearing in \cite[Theorem 3.3]{DLR}, it was crucial in \cite{CR} to assume that the levels of the three modular forms $f$, $g$ and $h$ were all equal, as otherwise the computations became too daunting. In the setting under study here, this assumption is not feasible because $p$ divides the level of $f$ but not the levels of $g$ and $h$. 


\end{enumerate}

In order to overcome these difficulties we add an extra $p$-adic variable to the computations of \cite[Ch.\,2]{DLR} by allowing the modular form $f$ to vary in a Hida family. While in loc.\,cit.\,$f$ was fixed throughout the article, here we adopt the strategy of regarding it as a member of a Hida family $\hf$ because the higher weight specialisations $\hf_k$, $k>2$, are (as opposite to $f$ itself) old at $p$. This allows us to work essentially in a setting where $p$ is a prime that behaves "as if it does not divide the level" and thus the necessary computations can then be performed more easily. 

In the process of our proof, we prove in Theorem \ref{factorization3} a factorisation formula of two-variable $p$-adic $L$-functions involving Castell\`{a}'s recent generalisation of \cite{BDP1} that the reader may find of independent interest. Just as the analogous one-variable formula proved in \cite[Theorem 3.9]{DLR} has found recently striking applications to the arithmetic of elliptic curves, we expect that our two-variable formula shall be also useful in this direction: cf.\,the forthcoming work of Bertolini and Darmon on Perrin-Riou's conjecture \cite{PR} for elliptic curves and Castell\`{a}'s work in progress on the Iwasawa theory of elliptic curves over imaginary quadratic fields.

 
	
\medskip\medskip\noindent
{\small
 \thanks{{\bf Acknowledgements.} The two authors were supported by Spanish Grant MTM2015-63829-P. The first author acknowledges financial support by Spanish Ministry of Economy and Competitiveness, through the Severo Ochoa programme for Centres of Excellence in R$\&$D (SEV-2015-0554). The second author acknowledges financial support from the Spanish Ministry of Economy and Competitiveness, through the Mar\'ia de Maeztu Programme for Units of Excellence in R$\&$D (MDM-2014-0445). This project has received funding from the European Research Council (ERC) under the European
Union's Horizon 2020 research and innovation programme (grant agreement No 682152). The authors also acknowledges financial support from the Spanish Ministry of Economy and Competitiveness, through the {\em Mar\'ia de Maeztu} Programme for Units of Excellence in R$\&$D (MDM-2014-0445).

}}

\medskip
\medskip




%==========GARRETT-HIDA============
\section{Garrett-Hida $p$-adic $L$-function}\label{Garrett}

The starting point in the proof of Theorem \ref{main-theta} is the interpolation formula satisfied by the Garrett-Hida $p$-adic $L$-function associated with a triple of Hida families passing through $f$, $g$ and $h$, that we now recall. 

Fix a rational prime $p\geq 3$ and set $\Gamma = 1+p\Zp$ and let $\Lambda = \Z_p[[\Gamma]]$ denote the Iwasawa algebra associated with $\Gamma$.

Define the formal scheme of weights as $\cX = \cX_{\Lambda} := \Spf(\Lambda)$. For any complete $\Z_p$-algebra $R$, the set of $R$-valued points of $\cX$ may be identified with
$$
\cX(R) = \Hom_{\text{cont}}(\Lambda, R) \simeq \Hom(\Gamma, R^\times).$$ 

As usual, the set $\Z_{\geq 2}$ can be embedded in $\cX(\Z_p)$ by identifying an integer $k$ with the character $\nu_k: \Gamma \lra \Z_p^\times$ sending $z$ to $z^{k}$. We let $\cX^{\cl} $ denote the set of such classical points.
%\[
%	\cX_F^{\cl} := \{ \nu_k = (\gamma \mapsto \gamma^k) \mid k \in \Z_{\geq 2} \} \subset %\cX_F(\cO_F)
%\]

Given a finite flat extension $\tilde \Lambda$ of $\Lambda$, there is  a natural projection $\cX_{\tilde\Lambda} \to \cX$ induced by the inclusion $\Lambda \hookrightarrow \tilde\Lambda$ and we define $\cX_{\tilde\Lambda}^\cl$ as the pull-back of $\cX^\cl$ under this map. The set $\cX_{\tilde\Lambda}^\cl$ is called set of {\em arithmetic primes} of $\tilde\Lambda$. By definition, any $\nu\in \cX_{\tilde\Lambda}^\cl$ lies above $\nu_k\in \cX^{\cl}$ for some $k=k(\nu)\geq 2$, which we refer to as the weight of $\nu$.

\begin{definition}
	Let $N_f\geq 1$ be an arbitrary integer not divisible by $p$ and $\chi$ be a Dirichlet character modulo $N_f$. A {\em Hida family} of tame level $N_f$ and tame character $\chi$ is a quadruple $(\Lambda_f, \cU_f, \cU_f^\cl, \hf)$ where
\begin{itemize} 

\item $\Lambda_f$  is a finite flat extension of $\Lambda$, 

\item $\cU_f\subset \cX_{\Lambda_f}$ is a rigid-analytic open subvariety, 

\item $\cU_f^\cl\subset \cU_f(\C_p)$ is a dense subset for the rigid analytic topology, and 

\item $\hf = \sum \ha_n q^n \in \Lambda_f[[q]]$ is a formal $q$-series such that for all $\nu\in \cU_f^\cl$ of weight $k = k(\nu)$,
	\[
		\nu(\hf) = \sum_{n\geq 1} \nu(\ha_n) q^n \in S_k(pN_f , \chi\omega^{k_0-k})
	\]
	is the $q$-expansion of an ordinary eigenform of weight $k$, level $pN_f$ and nebentype $\chi\omega^{k_0-k}$, for some $k_0 \in \{0, \dots, p-2 \}$. 

\end{itemize}
\end{definition}

Here $\omega: (\Z/p\Z)^{\times} \to \mu_{p-1}$ stands for the Teichm\"uller character of conductor $p$.

\begin{remark} In the literature one often allows a richer plethora of classical specializations, including twists by characters of $p$-power conductor of $\G$. For our purposes it suffices to restrict to the case in which this character is trivial. Given an arithmetic prime $\nu \in \cU_f^\cl$ over the classical point $\nu_k\in \cX$, it will often be harmless to abuse notation and write $\hf_k$ instead of $\nu(\hf)$.
\end{remark}

\begin{remark}\label{remark-weights}
 In order to simplify further the notations, we shall always restrict to those $k\in \Z_{\geq 1}$ for which $k \equiv k_0 \pmod{p-1}$, so that the nebentype character of $\hf_k$ will be exactly $\chi$. Under this condition, the specialisation $\hf_k$ of the Hida family $\hf$ at a point of weight $k>2$ is a classical eigenform which is always old at $p$ (cf.\,e.g.\,\cite[Lemma 2.1.5]{How}) and there exists an eigenform $f_k$ of level $N$ such that $\hf_k(q) = f_k(q)- \beta_{f_k} f_k(q^p)$ is the ordinary $p$-stabilization of $f_k$. Here $\beta_{f_k}$ is the single non-unit root of the Hecke polynomial $T^2-a_p(f_k) T + \chi(p) p^{k-1}$. The Hecke operator $U_p$ acts on $\hf_k$ with eigenvalue $\alpha_{f_k}$, the single $p$-adic unit root of the above polynomial.


If $k=2$, then $\hf_2$ is a classical eigenform that may either be old or new $p$. In the former case there exists as above an eigenform $f_2$ of level $N$ such that $\hf_2$ is the ordinary $p$-stabilisation of $f_2$. In the latter case we simply set $f_2 = \hf_2 $ and $\alpha_{f_2} = a_p(f_2)$, $\beta_{f_2}=0$. 

If $k=1$, the $p$-adic modular form $\hf_1$ may be classical or not: see \cite{DG} for a discussion of this phenomenon.
\end{remark}

Recall now the three classical eigenforms 
$$
f \in S_2(p N_f), \quad g\in M_1(N_g,\bar\chi), \quad h\in M_1(N_h,\chi)
$$ 
introduced at the beginning of this note. Choose eigenforms 
$$
\testf \in S_2(Np)[f], \quad \testg \in  M_1(N,\bar\chi)[g], \quad \testh \in M_1(N,\chi)_L[h]
$$
with respect to the good Hecke operators, having the same eigenvalues as $f$, $g$ and $h$ respectively.


Since $g$ and $h$ have weight $1$, the roots of the Hecke polynomials
$$
T^2-a_p(g)T+\bar\chi(p) = (T-\alpha_g)(T-\beta_g), \quad T^2-a_p(h)T+\chi(p) = (T-\alpha_h)(T-\beta_h)
$$
are all roots of unity. Fix arbitrary orderings $(\alpha_g,\beta_g)$, $(\alpha_h,\beta_h)$ of these roots and set
\begin{eqnarray*}
g_\alpha(q) := g(q)- \beta_g g(q^p), & h_\alpha(q) := h(q)-\beta_h h(q^p), \\ \testg_\alpha(q) := \testg(q)- \beta_g \testg(q^p), & \testh_\alpha(q) := \testh(q)-\beta_h \testh(q^p).
\end{eqnarray*}

Theorems of Hida \cite{Hi86} and Wiles \cite{W} ensure that there exist Hida families $\htf, \htg, \hth$ of tame level $N$ such that, for suitable arithmetic primes of weights $2$, $1$ and $1$ we have 
$$\htf_2(q) = \testf(q), \quad \htg_1(q) = \testg_\alpha(q), \quad \hth_1(q) = \testh_\alpha(q).$$


Define a dense subset of $\cU_f \times \cU_g \times \cU_h$ as follows:
\[
\cU_{fgh}^\cl := \{(k, \ell, m) \in \cU_f^\cl \times \cU_g^\cl \times \cU_h^\cl  \mid \ell \geq k+m \}.
\]
In \cite{DR1}, the authors constructed a three-variable $p$-adic $L$-function on $\cU_f \times \cU_g \times \cU_h$ which is determined by an interpolation formula relating the values at points in $\cU_{fgh}^\cl$ to central critical values of classical $L$-series: we refer to  \cite[prop. 2.3]{DLR} for full-fledged details. 

For our purposes it suffices to restrict the above $p$-adic $L$-function to  $\cU_{f g} := \cU_f \times \cU_g$, regarded as a rigid-analytic sub-variety of $\cU_f \times \cU_g \times \cU_h$ by setting the third variable to be the fixed point of weight one giving rise to $\testh_\alpha$. Set also $
\cU_{f g}^\cl := \{(k, \ell) \in \cU_f^\cl \times \cU_g^\cl, \ell \geq k+1 \}.$


With these notations, the constructions of \cite{DR1} and \cite[Ch.\,2]{DLR} give rise to a two-variable $p$-adic $L$-function 
$$
\Lp^g(\testhf, \testhg, \testh): \cU_{f g} \lra \C_p
$$
satisfying the following interpolation property for all $(k,\ell)\in \cU_{fg}^\cl$:
\begin{align} \label{DLRinterpolation}
	\Lp^g(\testhf, \testhg, \testh)\cdot \Lp^g (\testhf^*, \testhg^*, \testh^*)(k,\ell) &=  \gee_{\mathrm{GH}}(k,\ell)^2 \cdot \ga_\mathrm{GH}(k,\ell) \cdot \gf_\mathrm{GH}(k,\ell) \cdot \notag \\
	&\times \prod_{v\mid N\infty} C_v(\testf_k, \testg_\ell, \testh) \cdot  \frac{L(f_k\otimes g_\ell\otimes h, (k+\ell-1)/2 )}{\langle g_\ell^*, g_\ell^*\rangle ^2}
\end{align}
where:
\begin{itemize}
	\item For any modular form or Hida family $\phi$ of character $\chi$, we set $\phi^* = \phi \otimes \bar\chi$;

\vspace{0.1cm}

	\item $L(f_k,\otimes g_\ell\otimes h,s)$ is Garrett's triple-product complex $L$-function;

\vspace{0.1cm}

	\item
		$\ga_\mathrm{GH}(k,\ell) = (\frac{\ell+k-3}{2})!^2 (\frac{\ell-k-1}{2})!^2 \cdot \pi^{-2\ell}$;


\vspace{0.1cm}

	\item
		$\gee_\mathrm{GH}(k,\ell) =\cE(g_\ell, f_k, h)/\cE_0(g_\ell)\cE_1(g_\ell) = \cE( g^*_\ell, f^*_k, h^*)/\cE_0(g_\ell^*)\cE_1(g_\ell^*)$ is the Euler factor introduced in \cite[Theorem 1.3]{DR1};


\vspace{0.1cm}

	\item
		$\gf_\mathrm{GH}(k,\ell) = A\cdot 2^B$ where $A = A(N) \in \Q^\times$ and $B=B(k,\ell)$ is linear in $k,\ell$;


\vspace{0.1cm}

	\item $C_v(\testf_k, \testg_\ell, \testh)$ is a constant in the field generated by the fourier coefficients of $\testf_k$, $\testg_\ell$ and $\testh$; thanks to the work of Watson and Woodbury \cite{Wood}, when the three levels of the modular forms are square-free there exists a choice of test vectors $(\testf_k, \testg_\ell, \testh)$ such that the constants 
	\begin{equation}\label{Wv}
	W_v := C_v(\testf_k, \testg_\ell, \testh)
	\end{equation}
satisfy that  $W_v$ lies in $\Q^\times$  for all $v\mid N$ and only depends on $N$ and $v$ but not on the weights $(k,\ell)$ (see \cite[remark 2.2 and remark 2.4]{DLR}).
\end{itemize}

When we evaluate this function at the point $(2,1)\in \cU_{f g}$, which lies outside the region of classical interpolation, \cite[proposition 2.6]{DLR} asserts that:
\begin{equation} \label{DLRspecialvalue}
	\Lp^g(\htf, \htg, \testh)(2,1) = \int_{\testgamma_\alpha}\testf_2 \cdot \testh
\end{equation}
for a suitable choice of linear functional $\testgamma_\alpha : 	S_1(Np, \chi)_L[\testg^*_\alpha] \to L$, with $L=\Q(f_2, g_1, h)$.

%\begin{remark}
%	Notice that in our case $f_2 = \hf_2$, hence $p$ divides the level of $f$. Formula %\eqref{DLRspecialvalue} continue to hold in our setting since:
%	\[
%		\frac{\langle g_\ell,\phi_k\rangle_N}{\langle g_\ell,g_\ell\rangle_N} = \frac{\langle g_\ell,%\psi_k \rangle_{Np}}{\langle g_\ell,g_\ell\rangle_{Np}}
%	\]
%	hence the functional defined in \cite[prop. 2.6]{DLR} is well-defined also in our case when $k=2$.
%\end{remark}





%=============CASTELLA=================
\section{Castell\`{a}'s two-variable $p$-adic $L$-function}


Consider the Hida family $\hf$ as in the previous section. For any weight $\ka \in \cU_f^{\cl}$ such that $\ka \equiv 2 \,(\mbox{mod}\,\, p-1)$ let $f_\ka$ denote the  modular form of weight $\ka$ whose $p$-stabilization equals the weight $\ka$ specialization of $\hf$, as in Remark \ref{remark-weights}.

Let $K=\Q(\sqrt{-D_K})$, $D_K \geq 7$, be an imaginary quadratic field. We keep the same assumptions as in the introduction, so in particular we assume the Heegner hypothesis and we let $p=\wp \bar\wp$ be an odd prime that splits in $K$.

A Hecke character $\psi$ of $K$ of infinity type $(\ka_1,\ka_2)\in \Z \times \Z$ is a continuous homomorphism
$$
\psi: \mathbb{A}_K^\times \, \lra \, \C^\times
$$
satisfying $\psi(\alpha \cdot x \cdot z_\infty) = \psi(x) z_\infty^{\ka_1} \bar{z}_{\infty}^{\ka_2}$ for every $\alpha \in K^\times$, $x\in \mathbb{A}_K^\times$ and $z_\infty \in K_\infty^\times=\C^\times$. With this convention, the Hecke character  $\mathbf{N}_K$  induced by the norm from $K$ to $\Q$ has infinity type $(1,1)$. A Hecke character $\psi$ of infinity type $(\ka_1,\ka_2)$ is called {\em anticyclotomic} if its central character satisfies $\psi_{|\mathbb{A}_{\Q}^\times} = \mathbf{N}_K^{\ka_1 + \ka_2}$.

Let $\fc \subseteq \cO_K$ be an integral ideal and set $N=\mathrm{lcm}(N_f,D_K \Norm_{K/\Q}(\fc))$. Let $\Sigma_{\mathrm{cc}}$ denote the set of anticyclotomic Hecke characters of $K$ of conductor dividing $N$. For any Hecke character $\psi\in \Sigma_{\mathrm{cc}}$ of infinity type $(\ka_1,\ka_2)$ such that $\kappa = \ka_1 + \ka_2 \in \cU_f^{\cl}$, let $L(f_\ka,\psi,s)$ denote the $L$-function associated with the twist of $f_\ka$ by $\psi$, i.e.\, the $L$-function attached to the  compatible system of Galois representations afforded by the tensor product  $\varrho_{f_\ka|G_K} \otimes \psi$ of the (restriction to $G_K$ of) the Galois representations attached to $f_\ka$ and the character $\psi$. 



As usual, $L(f_\ka,\psi,s) = \prod_q L^{(q)}(q^{-s})$ is defined as a product of Euler factors ranging over the set of prime numbers. The Euler factors at the primes $q$ such that $q\nmid N$ are the same as that of the Rankin L-series $L(\theta_\psi \otimes f_\ka,s)$, but may differ at the primes $q$ such that $q\mid N$. Since $\psi$ is anticyclotomic, $s=0$ is the central critical point of the functional equation satisfied by $L(f_\ka,\psi^{-1},s)$.

For a given $\ka \in \cU_f^{\cl}$, let $\Sigma_{\ka,\mathrm{cc}}\subset \Sigma_{\mathrm{cc}}$ be the subset of anticyclotomic Hecke characters in $\Sigma_{\mathrm{cc}}$ of infinity type $(\ka_1,\ka_2)$ with $\ka_1 + \ka_2 = \ka$. This set is naturally the disjoint union of the three subsets 
$$
\Sigma^{(1)}_{\ka,\mathrm{cc}} = \{ \psi \in \Sigma_{\ka,\mathrm{cc}} \,\mbox{ of infinity type }\,(1,1)\},
$$
$$
\Sigma^{(2)}_{\ka,\mathrm{cc}} = \{ \psi \in \Sigma_{\ka,\mathrm{cc}} \,\mbox{ of infinity type }\,(\ka+\lam,-\lam), \lam \in \Z_{\geq 0}\}
$$
and
$$
\Sigma^{(2')}_{\ka,\mathrm{cc}} = \{ \psi \in \Sigma_{\ka,\mathrm{cc}} \,\mbox{ of infinity type }\,(-\lam,\lam+\ka), \lam \in \Z_{\geq 0}\}.
$$
  
Each of the three sets $\Sigma^{(1)}_{\ka,\mathrm{cc}}$, $\Sigma^{(2)}_{\ka,\mathrm{cc}}$ and $\Sigma^{(2')}_{\ka,\mathrm{cc}}$ is dense in the completion $\hat{\Sigma}_{\ka,\mathrm{cc}}$ of  $\Sigma_{\ka,\mathrm{cc}}$ with respect to the $p$-adic compact open topology as explained in \cite[\S 5.2]{BDP1}. In loc.\,cit.\,the authors construct a $p$-adic $L$-function on $\hat{\Sigma}_{\ka,\mathrm{cc}}$, denoted $\Lp(f_\ka/K)$, interpolating the algebraic parts of the central critical values of $L(f_\ka,\psi^{-1},s)$ at characters $\psi$ in $\Sigma^{(2)}_{\ka,\mathrm{cc}}$.

%As shown in \cite{BDP1}, there exists a unique $p$-adic analytic function
%\begin{equation*}
%	\Lp(f/K): \hat{\Sigma}_{f,K} \lra \C_p
%\end{equation*}
%interpolating the critical values $L(f,\psi^{-1},0)$ for $\psi \in \Sigma^{(2)}_{f,K}$, suitably normalized.

%We refer to $\Lp(f/K)$ as the Bertolini-Darmon-Prasanna $p$-adic Rankin $L$-function attached to the pair $(f,K)$. 

In \cite[\S 1]{Cas1}, Castell\`{a} constructs a two-variable $p$-adic $L$-function $\Lp(\hf/K)(\kappa, \psi)$ on the completion $\hat\Sigma_{\mathrm{cc}}$ of $\Sigma_{\mathrm{cc}}$, lifting the one-variable $p$-adic $L$-function of Bertolini-Darmon-Prasanna.  To state this more precisely, note that $\hat\Sigma_{\mathrm{cc}}$ is naturally fibered over $\cU_f$ and the fiber at a given point $\ka \in \cU_f^{\cl}$ is $\hat{\Sigma}_{\ka,\mathrm{cc}}$. Then for any $\ka \in \cU_f^\cl$ we have
\[
	\Lp(\hf/K)(\ka,-)^2 = \Lp(f_\ka/K). 
\]
%To genuinely and precisely express $\Lp(\hf/K)$ as a two variable $p$-adic L-function one has to be careful (see loc. cit. for further details). However, the convention we have chosen here is a little abuse of notation which simplify the treatment but do not affect the rigor of our proof.

Castell\`{a}'s $p$-adic L-function is characterized by the following interpolation property: let $\ka \in \cU_f$ and let $\psi \in \Sigma^{(2)}_{\ka,\mathrm{cc}}$ be a character of infinity type $(\ka+\lam, -\lam)$. Then \cite[Theorem 1.4]{Cas1} asserts that


%Recall the unramified Hecke characters $\psi_k$ introduced in \eqref{hecke-chars2}. It will be sufficient for our purposes to consider the restriction of $L_p(f,K)$ to the line arising as the $p$-adic closure of the set of Hecke characters 
%\begin{equation*}
%\{ \Psi_k:=\psi^{-1}_{2+2k} \mathbf{N}_K^{2+k}  \}
%\end{equation*}
%of infinity type $(0,-2-2k)+(2+k,2+k) = (2+k,-k)$. For $k\geq 0$ the character $\Psi_k \in \Sigma^{(2)}_{f,K}(\mathfrak N)$ lies in the region of interpolation of $L_p(f,K)$ and according to \cite[\S 5.2]{BDP1} the interpolation formula reads

 \begin{equation}\label{BDPinterpolation}
	\Lp(\hf/K)(\ka, \psi)^2 =\mathfrak a_{\mathrm{BDP}}(\psi) \cdot  \mathfrak{e}_{\mathrm{BDP}}(\ka, \psi)^2  \cdot \mathfrak{f}_{\mathrm{BDP}}(\ka, \psi)\cdot \frac{ \Omega_p^{2\ka+4\lam}  }{\Omega^{2\ka+4\lam}} \cdot L(f_\ka,\psi^{-1},0),
\end{equation} 
%{\color{blue}  Again, instead of $L(f,\Psi^{-1},0),$ you perhaps have $L(f_\alpha,\Psi^{-1},0)$ and $\omega(f_\alpha, \Psi)$ and this might be related to the problems you report in lemma 5.3 and remark 5.4. Of course in BDP the term $\omega(f_\alpha, \Psi)$ is only defined when $p$ does not divide the level, so one needs to do some work here. In \cite[Theorem 1.4]{Ca1} is unfortunately too vague, and the proof is not very good, because he decided to hide everything in his mysterious constant $C$.}

where

\begin{itemize}
\item
	$\mathfrak a_{\mathrm{BDP}}(\psi) = \lam!(\ka+\lam-1)! \pi^{\ka+2\lam-1}$
\item
$\mathfrak{e}_{\mathrm{BDP}}(\ka, \psi) = (1-\alpha_{f_\ka} \psi^{-1}(\bar\wp))(1-\beta_{f_\ka} \psi^{-1}(\bar\wp))  $,
\item
	$\mathfrak{f}_{\mathrm{BDP}}(\ka,\psi)  = \left(\frac{2}{c\sqrt{D_K}}\right)^{\ka+2\lam-1} \cdot 2^{\sharp \{ q\mid (D_K,N_E)\}} \cdot \omega(f_\ka,\psi)^{-1}$ 
\end{itemize}
with $\omega(f_\ka, \psi)$ as defined in \cite[(5.1.11)]{BDP1}.


If $\psi$ is a finite order anticyclotomic character of conductor $c\mid \mathfrak c$, then $\psi^{-1} \mathbf{N}_K$ is a character of infinity type $(1,1)$ and hence lies outside the region of interpolation. The main theorems of \cite{BDP1} and \cite[Theorem 2.10]{Cas2} assert that for such $\psi$ the following formula holds:
\begin{equation}\label{BDPspecialvalue}
	\Lp(\hf/K)(2, \psi \mathbf{N}_K) = \pm \mathfrak f_p(f,\psi) \cdot \log_{\omega_f}(D_{\psi})
\end{equation}
where $\mathfrak f_p(f,\psi) = (1- \alpha_{f_2} \psi^{-1}(\bar\wp) p^{-1})(1- \beta_{f_2}\psi^{-1}(\bar\wp) p^{-1})$ and $D_\psi$ is the class in $\Jac(X)$ of the divisor introduced in \eqref{eqn:choice-of-P}. 

Since $\pi_E^*(\omega_E) = c_E\cdot \omega_f$ and $P_\psi=\pi_E(D_\psi)$, it follows that
$$
 \log_{\omega_f}(D_{\psi}) = \frac{1}{c_E}\log_{\pi_E^*(\omega_E)}(D_\psi) = \frac{1}{c_E}\log_{E,p}(P_\psi),
$$
and hence
\begin{equation}\label{BDPspecialvalue-bis}
	\Lp(\hf/K)(2, \psi \mathbf{N}_K) = \pm \mathfrak f_p(f,\psi) \cdot \frac{1}{c_E}\log_{E,p}(P_\psi).
\end{equation}


%==============KATZ==================
\section{Katz p-adic $L$-function}

Keep the notations and assumptions of the previous section. Denote by $\Sigma_K$ the set of Hecke characters of $K$ of conductor dividing $\fc$ and define 
%$$\Sigma_K = \Sigma_K^{(1)}\cup \Sigma_K^{(2)}\subset \Sigma$$ to be the disjoint union of the sets
$$
\Sigma_K^{(1)}=\{ \psi \in \Sigma \mbox{ of infinity type } (\ka_1,\ka_2), \ka_1\leq 0, \ka_2\geq 1\},
$$
$$
\Sigma_K^{(2)}=\{ \psi \in \Sigma \mbox{ of infinity type } (\ka_1,\ka_2), \ka_1\geq 1, \ka_2\leq 0\}.
$$

For all $\psi \in \Sigma_K^{(2)}$, $s=0$ is a critical point for the Hecke $L$-function $L(\psi^{-1},s)$, and Katz's $p$-adic $L$-function is constructed by interpolating the (suitably normalized) values $L(\psi^{-1},0)$ as $\psi$ ranges over $\Sigma_K^{(2)}$. 

More precisely, let $\hat{\Sigma}_K$ denote the completion of $\Sigma_K^{(2)}$ with respect to the compact open topology on the space of functions on a certain subset of $\mathbf{A}^\times_K$, as described in \cite[\S 5.2]{BDP1}. By the work of Katz \cite{Katz1}, there exists a $p$-adic analytic function
$$
\Lp(K): \hat{\Sigma}_K\lra \C_p
$$
which is characterized by the following interpolation property: for all $\psi \in \Sigma_K^{(2)}$ of infinity
%\footnote{In \cite[Thm.\,4.14]{dS}, he in fact asks the weights to satisfy $\ka_1>|\ka_2|$. (In his notation, $\ka_1=\ka$ and $\ka_2=j$.) This restriction is %not imposed in BDP for some reason...} 
type $(\ka_1,\ka_2)$,
\begin{equation}\label{Katzinterpolation}
	\Lp(K)(\psi) = \gak(\psi) \cdot \gek(\psi) \cdot  \gfk(\psi) \frac{\Omega_p^{\ka_1-\ka_2}}{\Omega^{\ka_1-\ka_2}}L_\fc (\psi^{-1},0)
\end{equation}
where
\begin{itemize}
\item $L_\fc (\psi^{-1},s)$ is Hecke's $L$-function associated with $\psi^{-1}$ with the Euler factors at primes dividing $\fc$ removed,

\item $\Omega_p\in \C_p^\times$ is a $p$-adic period attached to $K$, as defined in \cite[(140)]{BDP1},

\item $\Omega\in \C^\times$ is the complex period associated with $K$ as defined in \cite[(137)]{BDP1},

\item
	$\gak(\psi) = \frac{(\ka_1-1)!}{\pi^{\ka_2}}$
\item
	$\gek(\psi) = (1-\frac{\psi(\wp)}{p}) (1-\psi^{-1}(\bar\wp))$
\item
	$\gfk(\psi) = \left( \frac{\sqrt{D_K}}{2} \right)^{\ka_2}$.

\end{itemize}


%In particular, if the character $\psi$ is anticyclotomic (as is it in our case), then $L_p(K)(\psi) = L_p(K)(\psi\mathbf{N}_{K})$. 

%Indeed, recall that Gross' $p$-adic $L$-function differs from ours: up to a constant, $L_p^{Gr}(\psi) = L_p(K)(\psi^{-1})$. This implies that while $L_p^{Gr}$ satisfies a functional equation relating $\psi$ to $\hat\psi = (\psi')^{-1}\mathbf{N}_K^{-1} = \psi \mathbf{N}_K^{-1}$, our $p$-adic $L$-function relates then $\psi$ to $\psi \mathbf{N}_K$.

%The values of a $p$-adic $L$-function at points lying outside the region of interpolation are of great interest to us, as they may encode arithmetic invariants of the motives under consideration. One instance of this philosophy is %

The following result is commonly known as Katz's Kronecker $p$-adic limit formula. It computes the value of $\Lp(K)$ at a finite order character $\psi$ of $G_K$, which lies outside the region of interpolation (cf.\,\cite[\S 10.4, 10.5]{Katz1}): 
\begin{equation}\label{KatzKronecker}
	L_p(K)(\psi) = \mathfrak f_p(\psi) \cdot \log_p(u_{\psi}),
\end{equation}
where
\begin{equation}\label{Katz-GZ}
	\mathfrak f_p(\psi) % =  L_p(K)(\bar{\psi}_g\mathbf{N}_{K}) 
= \begin{cases}\frac{1}{2} (\frac{1}{p}-1)  & \mbox{ if } \psi=1 \\
 \frac{-1}{24c}  (1-\psi(\bar\wp)) (1-\frac{\psi(\bar\wp)}{p}) &  \mbox{ if } \psi\ne 1.
\end{cases}
\end{equation}
Here $c>0$ is the smallest positive integer in the conductor ideal of $\psi$.









\section{Proof of Theorem \ref{main-theta}}\label{proof1}

%==============ACTUAL PROOF=============

Let
\[
f\in S_2(p N_f), \qquad	g=\theta_{\psi_g} \in M_1(D_K N(\fc_g), \barchi),  \qquad  h=\theta_{\psi_h} \in M_1(D_K N(\fc_h), \chi)
\]
be the three modular forms considered in the introduction, and we keep the assumptions as therein. Remember that $p = \wp\overline{\wp}$ is a prime that splits in $\cO_K$ and does not divide $c N_f$.

Let $\hf$ and $\hg$ be cuspidal Hida families passing through $f$ and $g_\alpha$, that is to say, satisfying $\hf_2 = f$ and $\hg_1=g_\alpha$. The Hida family $\hf$ is unique thanks to the classical work of Hida \cite{Hi86}. The Hida family $\hg$ is also unique thanks to the classicality hypothesis assumed in the introduction, by a recent result of Bellaiche and Dimitrov \cite{BeDi}. This family is in fact CM and its higher weight classical specializations admit an explicit description that we now review.

Take a Hecke character $\lambda$ with image in $\Z_p^\times$ of infinity type $(0,1)$ and conductor $\bar\wp$. Following \cite[p. 235-236]{Hi93}, with a slight modification, we define the Hecke characters $\psi_{g,\ell-1}$, of conductor $\mathfrak c_g$, as follows: 
\begin{align*}
	\psi_{g, \ell-1}(\mathfrak q) &:= \psi_g (\mathfrak q) \langle \lambda \rangle^{\ell-1}(\mathfrak q), \qquad \text{ for } \mathfrak q \ne \bar\wp \\
	\psi_{g, \ell-1}(\bar\wp) &:= \barchi(p) \frac{p^{\ell-1}}{\psi_{g, \ell-1}(\wp)}.
\end{align*}

For every $\ell \in \cU_g^{\cl}$ one then has that $\hg_\ell$ is the ordinary $p$-stabilisation of $g_\ell := \theta_{\psi_{g, \ell-1}}$. Using the notations introduced above, note that we have
\[
	\alpha_{g_\ell} = \psi_{g, \ell-1}(\wp), \qquad \beta_{g_\ell} = \psi_{g, \ell-1}(\bar\wp).
\]

Recall the three ring class characters $\psi_0$, $\psi_1$, $\psi_2$ defined in the introduction. 
Recall also that $\mathbf N_K$ stands for the Hecke character of infinity type $(1,1)$ induced by the norm from $K$ to $\Q$. 
Associated with them there are three additional families of characters which play a key role in the proof:
\begin{align}
	\Phi_g(\ell) &:= (\psi_{g,\ell-1}^2\chi)^{-1} \mathbf N_K^\ell,   & &\text{for which } \Phi_g(1) = \psi_0' \mathbf N_K \label{Phi} \\
	\Psi_{gh}(k,\ell) &:= ({\psi}_{g,\ell-1}\psi_h)^{-1} \mathbf N_K^{\frac{k+\ell-1}{2}},   & &\text{for which } \Psi_{gh}(2,1) = \psi_1' \mathbf N_K \label{Psigh} \\
	\Psi_{gh'}(k,\ell) &:= (\psi_{g,\ell-1}{\psi'}_h)^{-1} \mathbf N_K^{\frac{k+\ell-1}{2}},   & &\text{for which } \Psi_{gh}(2,1) = \psi_2' \mathbf N_K \label{Psigh'}.
\end{align}

Notice that $\Phi_g(\ell)$ is of infinity type $(\ka_1, \ka_2) = (\ell, -\ell+2)$, while $\Psi_{gh}(k,\ell)$ and $\Psi_{gh'}(k,\ell)$ are of infinity type $(\ka+\lambda, -\lambda)$ with $\ka = k$ and $\lambda = (\ell-k-1)/2 \geq 0$ because $\ell \geq k+1$.

\begin{definition}\label{ad}
	Let $L$ be a number field. A function $$\mathfrak f : \cU_g^{\cl} \to \overline{\Q}$$ is $L$-admissible if it extends to a meromorphic Iwasawa function on $\cU_g$ having no pole at $1$ and satisfying $\mathfrak f(1) \in L^\times$. Similarly, a function
$$\mathfrak f : \cU_{fg}^{\cl} \to \overline{\Q}$$ is $L$-admissible if it extends to a meromorphic Iwasawa function on $\cU_{fg}$ having no pole at $(2,1)$ and satisfying $\mathfrak f(2,1) \in L^\times$.
\end{definition}

Set $\gap(\ell) =  \frac{(\ell-1)!}{\pi^\ell}$.
By \cite[Lemma 3.7 and (53)]{DLR} there exists a $\Q(\psi_g, \psi_h)$-admissible function $\gfp$ such that
\begin{equation} \label{Petersson}
	\langle g_\ell^*, g_\ell^* \rangle  = \gap(\ell) \cdot \gfp(\ell) \cdot L(\Phi_g(\ell)^{-1},0).
\end{equation}


\begin{lemma} \label{charactersatp} The following formulae hold true:
\begin{align*}
	\Phi_g(\ell)(\wp) &= \frac{\beta_{g_\ell}^2 \chi(p)}{p^{\ell-2}}, & \Phi_g(\ell)(\bar\wp) &= \frac{p^\ell }{\beta_{g_\ell}^2\chi(p)}, \\
	\Psi_{gh}(k,\ell)(\bar\wp) &= \frac{p^{\frac{\ell+k-1}{2}} }{\beta_{g_\ell}\beta_{h}}, & \Psi_{gh'}(k,\ell)(\bar\wp) &= \frac{p^{\frac{\ell+k-1}{2}} }{\beta_{g_\ell}\alpha_{h}}.
\end{align*}	
\end{lemma}

\begin{proof}
	This follows from a straight-forward computation.
\end{proof}

\begin{lemma} \label{simplify}
The following equalities hold true:
\begin{align*}
	 \gee_{K}(\Phi_g(\ell))^2 \cdot \geh(k,\ell)^2  & =  \gee_{\mathrm{BDP}}(k,\Psi_{gh}(k,\ell))^2 \gee_{\mathrm{BDP}}(k,\Psi_{gh'}(k,\ell))^2, \\
	  \gah(k,\ell) \cdot \gak(\Phi_g(\ell))^2  &= \gab(\Psi_{gh}(k,\ell)) \cdot \gab(\Psi_{gh'}(k,\ell)) \cdot \gap(\ell)^2.
 %\\
	% \frac{ \gfh(\ell,k,1) \cdot \gfk(\Phi_g(\ell)) }{ \gfb(\Psi_{gh}(k,\ell)) \cdot \gfb(\Psi_{gh'}(k,\ell)) }  &=
\end{align*}
\end{lemma}

\begin{proof}
This again follows from a straight-forward computation using Lemma \ref{charactersatp} for the first equality.
\end{proof}

Recall the factor $\omega(f, \Psi)$ appearing in \eqref{BDPinterpolation} and defined in \cite[equation (5.1.11)]{BDP1}. Recall also from the remarks around the Heegner hypothesis in the introduction the integral ideal $\mathfrak N$ of $\cO_K$ such that $\Norm_{K/\Q}(\mathfrak N) = N_f$.


\begin{lemma} \label{BDPomega} 
The following identities hold true:
	\[
		\omega(f, \Psi_{gh}(k,\ell)) = (-1)^{\frac{\ell-1}{2}} \frac{\psi_{g,\ell-1} (\mathfrak N) \psi_h(\mathfrak N)}{N_f^{\frac{\ell-1}{2}}}, \qquad \omega(f, \Psi_{gh}(k,\ell)) = (-1)^{\frac{\ell-1}{2}} \frac{\psi_{g,\ell-1} (\mathfrak N) \psi_h'(\mathfrak N)}{N_f^{\frac{\ell-1}{2}}}.
	\]
	In particular we have
	\[
		\omega(f, \Psi_{gh}(2,1)) = \psi_1(\mathfrak N) \quad \mbox{and} \quad \omega(f, \Psi_{gh}(2,1)) = \psi_2(\mathfrak N).
	\]
\end{lemma}

\begin{proof}
	We limit ourselves to compute $\omega(f, \Psi_{gh}(k,\ell))$, as the other case is very similar. For simplicity, call $\Psi := \Psi_{gh}(k,\ell)$ the Hecke character of infinity type $(\ka + \lam, - \lam)$, where $\ka = k$ and $\lam = (\ell-k-1)/2$. Define
	\[
		\Psi_\lam := \Psi   \mathbf N_K^\lam.
	\]
	Choose an ideal $\mathfrak b \subset \cO_c$ relatively prime to $N_fc$ and an element $b_N \in \cO_c$ such that $\mathfrak b \mathfrak N = (b_N)$. Since $\omega_f = 1$ and $\varepsilon_f = 1$ in our case, \cite[equation (5.1.11)]{BDP1} shows that
	\begin{equation} \label{definitionomega}
		\omega(f,\Psi) = \Psi_\lam(\mathfrak b) \cdot (-1)^{\ka/2 + \lam} N^{\ka/2 + \lam} b_N^{-\ka -2\lam}. 
	\end{equation}
	By  \cite[equation (5.1.2)]{BDP1} it then follows that
	\[
		\Psi_\lam(\mathfrak b) \Psi_\lam(\mathfrak N) = \Psi_\lam((b_N)) = b_N^{\ka + 2\lam}.
	\]
Substituting into \eqref{definitionomega} we find:
	\[
	\omega(f,\Psi) = (-1)^{\ka/2 + \lam}\frac{N_f^{\ka/2 + \lam}}{\Psi_\lam(\mathfrak N)}  = (-1)^{\ka/2 + \lam}\frac{N_f^{\ka/2}}{\Psi(\mathfrak N)}.
	\]
	Using now the definition of $\Psi_{gh}(k,\ell)$ and the fact that $\ka +2\lam = \ell-1$ we find
	\[
		\omega(f,\Psi) = (-1)^{\frac{\ell-1}{2}} \cdot \frac{\psi_{g,\ell-1}(\mathfrak N) \psi_h(\mathfrak N) N_f^{k/2}}{N_f^{\frac{\ell+k-1}{2}}} =  (-1)^{\frac{\ell-1}{2}} \cdot \frac{\psi_{g,\ell-1}(\mathfrak N) \psi_h (\mathfrak N)}{N_f^{\frac{\ell-1}{2}}},
	\]
as claimed.
%	In particular, for $\ell=1$ we get the desired result.
\end{proof}

%\begin{remark}
%	The above lemma shows that the factor $\omega(f,\Psi)$ is independent on the choice of $(\mathfrak b, b_n)$, as in fact proved in \cite[lemma 5.3]{BDP1}. Moreover, it only depends on the weight of $g$. This is happens because the infinity type of the Hecke character $\Psi_{gh}(k,\ell)$ is $(\ka + \lam, -\lam)$ where $\lam$ depends in a specific way by $\ka = k$.
	
%	Notice that $\psi_{g,l-1}$ take values which can be interpolated as $\ell$ varies. By definition it is a unit which varies analytically, except for $\bar\wp$, but $\bar\wp \nmid \mathfrak N$.
	
%	The point is: does $N^{\frac{\ell-1}{2}}$ extends analytically? If $p\nmid N$ it does, if $p\mid N$ it doesn't.
	
%	If we fix $k=2$, the problem is real! At least, it seems to be. When $k= 2$, $N=N_f$, so that $p\mid\mid N_f$, then the resulting function doesn't interpolate $p$-adically. 
	
%{\color{blue}  See my comments in blue in pages 4 and 5. I think there we should find the key to our problem.}
	
%\end{remark}



%\begin{remark}
%	The things I said in the above create problems. I think that the reason for which they do create a problem is the following: in the definition we adopted now of $\psi_{g,\ell-1}$, the value is a $p$-adic unit for every prime ideal different from $\bar\wp$. In particular, the $\omega(f, \Psi) \notin \cO_{K_\wp}^\times$ (or an extension).
	
%	The problem is that \cite{BDP1}, when introducing the scalar in equation $(5.1.11)$, they specify that it is a complex unit. Since it is algebraic, it must be a root of unity which goes to $\cO_{K_\wp}^\times$ (or an extension).
	
%	After noticing this, I agree that the value of $p$ should simplify in this computation. I localized the mistake in the definition of the character $\psi_{g,\ell-1}$. In \cite[pages 235-236]{Hi93}, Hida introduced the character $\lambda$ of infinity type $(1,0)$ and conductor $c\wp$. In our case I adopted the definition with conductor $\bar\wp$ to let us have $\alpha_{g_\ell} = \psi_{g,\ell-1}(\wp)$ and $\beta_{g_\ell} = \psi_{g,\ell-1}(\bar\wp)$. FALSE: I notice just now that what I've done might be wrong because by definition of the infinity type, for $\pi$ generator of the principal ideal $\wp$ in $K_\wp$ we want $\lambda((\pi))= \pi$ which is not a $p$-adic unit.
%\end{remark}



%{\bf Reword}. Starting from the definitions \eqref{Phi}, \eqref{Psigh} and \eqref{Psigh'}, it follows easily that the values of the functions $\Lp^{g_\alpha}(\hf,\hg, h)(k,\ell)$, $\Lp(\hf/K)%(\Psi_{gh}(k,\ell))$, $\Lp(\hf/K)(\Psi_{gh'}(k,\ell))$ and $\Lp(K)(\Phi_g(k))$ lie in the region of interpolation, hence they can be compared.

Let
$$
L_0=K(\psi_g, \psi_h, \tau(\chi), \sqrt{N_g}, \sqrt{N_h})
$$
denote the extension of $\Q$ generated by the values of the Hecke characters $\psi_g$ and $\psi_h$, the Gauss sum associated with the Dirichlet character $\chi$ and the square-roots of $N_g$ and $N_h$.

Fix test vectors $\htf$, $\htg$ and $\testh$ as in \S \ref{Garrett}, giving rise to the $p$-adic $L$-function $\Lp^g(\testhf, \testhg, \testh)$ satisfying the good properties discussed around \eqref{Wv}.

\begin{theorem} \label{factorization1} There exist a quadratic extension $L/L_0$ and a $L$-admissible function $\gf$ on  $\cU_{fg}$ such that the following factorization of two-variable $p$-adic $L$-functions holds:
	\begin{eqnarray*}
		\Lp^{g}(\htf,\htg,\testh)(k,\ell) \cdot  \Lp(K)(\Psi_{\hg}(\ell))  =  \qquad \qquad  \qquad \qquad  \qquad \qquad \qquad \qquad\\ \nonumber  \qquad \qquad \qquad \qquad  \gf(k,l) \cdot  \Lp(\hf/K)(k, \Psi_{\hg h}(k,\ell)) \, \cdot \,\Lp(\hf/K)(k, \Psi_{\hg h'}(k,\ell)).
	\end{eqnarray*}
\end{theorem}

\begin{proof}
We have a decomposition of classical $L$-functions given by
\begin{equation} \label{classicaldecomposition}
	L\Bigl(f_k\otimes g_\ell \otimes h, \frac{k+\ell-1}{2} \Bigr) = \cEul_N(k,\ell)\cdot L(f_k, \Psi_{gh}(k,\ell)^{-1}, 0)\cdot L(f_k, \Psi_{gh'}(k,\ell)^{-1}, 0).
\end{equation} 
Since $p\nmid N$, the Euler factor in $\cEul_N(k,\ell)$ interpolates $p$-adically and gives rise to a two-variable $L_0$-admissible function. Combine equations \eqref{Petersson} and \eqref{classicaldecomposition} with the interpolation formula \eqref{DLRinterpolation}. Then use equations \eqref{BDPinterpolation} and \eqref{Katzinterpolation} to replace the classical $L$-functions with their respective $p$-adic avatar. An easy check shows that the periods simplify. 

Define
\[
\omega(k,\ell) := I(\testg_\ell^*,\testf_k^*, h^*)/I(\testg_\ell, \testf_k, \testh)
\]
as in \cite[(25)]{DLR}. Recall also the local constants $W_v$ for $v\mid N \infty$ introduced in \eqref{Wv}.

Thanks to Lemma \ref{simplify} one obtains the following equality, true for every $(k,\ell) \in \cU_{f g}^{\cl}$:
\begin{eqnarray}\label{factor}
	\Lp^{g}(\testhf,\testhg,\testh)(k,l)^2 \cdot  \Lp(K)(\Phi_g(\ell))^2 = \qquad \qquad  \qquad \qquad  \qquad \qquad \qquad \qquad\\ \nonumber  \qquad \qquad \qquad \qquad \gf_0(k,\ell) \cdot \Lp(\hf/K)(k,\Psi_{gh}(k,\ell))^2 \cdot \Lp(\hf/K)(k,\Psi_{gh'}(k,\ell))^2
\end{eqnarray}
where
\[
	 \mathfrak{f}_0(k,\ell) = \frac{\cEul_N(k,\ell)}{\gfp(\ell)^2} \cdot \frac{ \gfh(\ell,k,1) \cdot \gfk(\Phi_g(\ell))^2 }{ \gfb(\Psi_{gh}(k,\ell)) \cdot \gfb(\Psi_{gh'}(k,\ell)) } \cdot \frac{ \prod_{v\mid N\infty} W_v }{\omega(k,\ell)}.
\]
	

%Since $\gf_0(k,\ell)$ is a quotient of $p$-adic $L$-functions it is a F-admissible function (can be extended to an Iwasawa function over some extension of the Iwasawa algebra). It remains to show that it is defined over $L_0$ and that it is non-zero at $(k.\ell)=(1,1)$.

\vspace{0.2cm}

Let us show that $\mathfrak f_0$ is $L_0$-admissible.  First of all, it is immediate to verify that the powers or $\pi$ appearing in the various fudge factors cancel out. Notice also that almost all terms appearing in the numerator and denominator of the definition of $ \mathfrak{f}_0(k,\ell)$ are $L_0$-admissible for obvious reasons, following directly from the definitions. This is the case except for the constants $W_v$, $\omega(f_k, \Psi_{gh})\omega(f_k, \Psi_{gh'})$ and $\omega(k,\ell)$, which need to be considered in more detail.

As for the local constants $W_v$ is concerned, we already argued in \S \ref{Garrett} that for $v\mid N$ the test vectors can be chosen in such a way that $W_v$ lies in $\Q^\times$ and do not depend on $(k,\ell)$. Moreover, the recent results of \cite{Wood2} guarantee that in this setting $W_\infty=1$ for all $(k,\ell)$. 
 
As for the global constant $\omega(f_k, \Psi_{gh})\omega(f_k, \Psi_{gh'})$, using lemma \ref{BDPomega} one derives that:
\[
	\omega(f_k,\Psi_{gh})\omega(f_k,\Psi_{gh'}) =  \frac{\psi_{g,\ell-1}(\mathfrak N)^2 \psi_h(\mathfrak N) \psi_h'(\mathfrak N)}{N_f^{l-1}}.
\]
Since $p\nmid N_f$, it follows that the function $\omega(f,\Psi_{gh})\omega(f,\Psi_{gh'})$ is $L_0$-admissible (in fact it is $K(\psi_g, \psi_h)$-admissible).

The $L_0$-admissibility of the function $\omega(k,l)$ follows by the same argument as in the last part of the proof of \cite[Theorem 3.9]{DLR}. Hence we have proved that $\mathfrak f_0$ is $L_0$-admissible and the theorem follows after taking the square-roots on both sides of \eqref{factor}.
\end{proof}




\begin{corollary}[Elliptic Stark conjecture] There exists a scalar $\lambda \in L^\times$ such that:
\[
	\int_{\testgamma_{g_\alpha}} \testf\cdot \testg = \lambda \cdot \frac{ \log_{E,p}(P_{\psi_1'}) \log_{E,p}(P_{\psi_2'})}{\log_p(u_{\psi_0'})}.
\]
\begin{proof}
	Notice that we have $$\Phi_g(1) = \psi_g'/\psi_g = \psi_0', \quad \Psi_{gh}(2,1) = \psi_1' \quad \mbox{and} \quad \Psi_{gh'}(2,1) = \psi_2'.$$ Using Theorem \ref{factorization1} and equations \eqref{DLRspecialvalue}, \eqref{BDPspecialvalue-bis} and \eqref{KatzKronecker} one obtains the desired result.
\end{proof}
\end{corollary}	






\section{An explicit formula}\label{sec:lambda}

Let us assume now that $\psi = \psi_g$ is a ring class character of conductor $c\in \Z_{\geq 1}$ and $\psi_h = 1$ is the trivial character. This implies that $h= \theta_{\psi_h} = E_{1, \chi_K}$ is the Eisenstein series associated with the pair of Dirichlet characters $(1,\chi_K)$. In this setting the characters defined in equations \eqref{Psigh} and \eqref{Psigh'} coincide, i.e. $\Psi_{gh} = \Psi_{gh'}$, and thus we simply denote this character $\Psi_g$. 

In this section we exploit the results obtained previously to prove an explicit formula for the scalar $\lambda$ that makes its appearance in Theorem \ref{main-theta}, in the particular scenario considered here. In order to do so we invoke yet another $p$-adic $L$-function in addition to the ones introduced so far. Namely, the {\em Hida-Rankin} $p$-adic $L$-function associated by Hida in Chapter 10 of  \cite{Hi93} to the convolution of two Hida families $\hf$, $\hg$ of cusp forms.

We follow the conventions and notations employed in \cite[\S 2]{BDR1} and \cite[\S 3]{CR}, where Hida-Rankin's $p$-adic $L$-function $\Lp^g(\hf,\hg)$ is described as a function on $\cU_{fg}$ and is characterized by the following interpolation property: for every $(k,\ell)\in \cU_{f g}^{\cl}$ such that $\ell \geq k+1$, it holds that
\begin{equation}\label{Hidainterpolation}
	\Lp^g(\hf,\hg)(k,\ell) = \ger(k,\ell)\cdot \gar(k,\ell) \cdot \gfr(k,\ell) \cdot \frac{L(\hf_k \otimes \hg_\ell, (\ell+k-1)/2)}{\langle \hg_\ell, \hg_\ell \rangle_N}
\end{equation}
where
\begin{itemize}
	\item
		$\ger(k,\ell) = \geh(k,\ell,1)$;
	\item
		$\gar(k,\ell) = N/\sqrt{D_K}$
	\item
		$
			\gfr(k,\ell) =  (-1)^{\frac{\ell-k-1}{2}} \cdot \Bigl( \frac{\ell-k-1}{2} \Bigr)! \Bigl(\frac{\ell-k-3}{2} \Bigr)! \cdot 2^{-2\ell+1} \pi^{-\ell}.
		$
\end{itemize}

Moreover, by \cite[Proposition 3.2]{CR} we dispose of the following formula for the value of $\Lp^g(\hf,\hg)$ at the point $(k,\ell)=(2,1)$ lying outside the region of interpolation. Keeping the notations of the introduction and section \ref{Garrett}, we have:
\begin{equation} \label{Hidaspecialvalue}
	\Lp^g(\hf,\hg)(2,1) = \int_{\testgamma_{g_\alpha}}\testf\cdot \testh.
\end{equation}


Recall  the number field appearing in the statement of Theorem \ref{factorization1}.

\begin{proposition} \label{factorization3}
	There exists a two-variable $L$-admissible function $\gf(k,\ell)$ such that the following factorization of $p$-adic $L$-functions holds:
	\[
		\Lp^g(\hf,\hg)(k,\ell) \cdot \Lp(K) (\Phi_g(\ell)) \, = \,  \gf(k,\ell) \cdot \Lp(\hf/K)(\Psi_{gh}(k,\ell))^2.
	\]
	\end{proposition}

\begin{proof} After setting $ \Psi := \Psi_{gh}(k,\ell)$, it is easy to verify that
\[
		L(\testg_\ell \otimes \testf_k, (\ell+k-1)/2) = \cEul_N(k,\ell) \cdot L(f_k, \Psi,0)
	\]
where $ \cEul_N(k,\ell)$ stands for a product of Euler factors at primes dividing $N$. The function $\cEul_N(k,\ell)$ gives rise to an admidsible function as shown in \cite[Lemma 4.1]{CR}. From this, the proof proceeds along very similar lines as in the proof of Theorem \ref{factorization1} and  \cite[Theorem 4.4]{CR}, replacing the interpolation formula of the Garrett-Hida triple-product $L$-function by the one provided above. 

We content here to record an explicit expression for the admissible function appearing in the above statement, namely
\begin{equation}\label{fff}
		\gf(k,\ell) = \cEul_N(k,\ell)\cdot (-1)^{\frac{k}{2}}\frac{\Im(D_Kc^2)}{\Im(N)} \frac{N\cdot 2^{-\sharp \{q\mid(D_K,N_f)\}}}{h_c \cdot D_K} \frac{\psi_{\ell-1}(\mathfrak N)}{c^{\ell+1}N_f^{\frac{\ell-1}{2}}}
\end{equation}
where for any integer $M\geq 1$ we set $\Im(M) :=[\mathrm{SL}_2(\Z): \Gamma_0(M)] = \prod_{q^{n_q}\mid \mid M} q^{n_q-1}(q+1)$.
\end{proof}



\begin{corollary}
Assume $N_f=D_K$ and $g=h = \mathrm{Eis}_1(1, \chi_K)$. Then
	\[
		\int_{\gamma_{g_\alpha}} f \cdot h = \lambda\cdot \frac{ \log_{E,p}^2 (P_{\psi'}) }{ \log_p(u_{\psi'^2}) }
	\]
	with
	\[
		\lambda = \frac{ (p-a_p(f) \psi(\wp))^2}{p \cdot c_E^2 }\cdot \frac{\lambda_0}{h_K g_K}
	\]
	where:
	\[
		\lambda_0 = \begin{cases} \frac{1}{p-1}   & \quad \mbox{if } \psi^2=1, \mbox{ that is to say, if } g \mbox{ is Eisenstein} \\
\frac{12}{p-(p+1)\psi^2(\bar \wp)+\psi^4(\bar \wp) } & \quad \mbox{if } \psi^2\ne 1, \mbox{ that is to say, if } g \mbox{ is cuspidal.}
\end{cases}	
	\]
\end{corollary}

\begin{proof}
The value of the admissible function $\mathfrak f$ appearing in \eqref{fff} at the point $(k,\ell)=(2,1)$ is
	\[
		\mathfrak f(2,1) = -\frac{2 \psi(\mathfrak N) }{h_Kg_K}.
	\]
	The result now follows upon combining this with formulae \eqref{BDPspecialvalue-bis}, \eqref{KatzKronecker} and \eqref{Hidaspecialvalue}.
\end{proof}






\end{document}
